Author's response to reviews

Title: The effects of a brief CBT intervention, delivered by frontline mental health staff, to promote recovery in people with psychosis and comorbid anxiety or depression (the GOALS study): study protocol for a randomized controlled trial.

Authors:

Helen Waller (helen.waller@kcl.ac.uk)
Tom Craig (thomas.craig@kcl.ac.uk)
Sabine Landau (sabine.landau@kcl.ac.uk)
Miriam Fornells-Ambrojo (miriam.fornells-ambrojo@ucl.ac.uk)
Nedah Hassanali (nedah.hassanali@slam.nhs.uk)
Catherine Iredale (catherine.iredale@slam.nhs.uk)
Suzanne Jolley (suzanne.jolley@kcl.ac.uk)
Paul McCrone (paul.mccrone@kcl.ac.uk)
Philippa Garety (philippa.garety@kcl.ac.uk)

Version: 4 Date: 5 June 2014

Author's response to reviews:

The title has now been revised to conform to the journal style.